vc/18-Nov-03
Week 47 starting 17 nov 03

SM18 Operation SCHEDULE Meeting (Tuesdays)

Schedule for this week : 
	BENCH
	CURRENT

MAGNET
	STATUS 

Today(18/11/03)
	Forward look,   & Next Magnet 

	TBA1
	NOT Active
	
	Bench de-activated – preparation for D bench

“Park” here 2001 ( moved from D bench )

	TBA2
	3022
	cold
	

	TBC1
	3027
	To warm up & disconnect
	1027 next ?

	TBC2
	SSS04
	cold
	

	TBE1
	2027
	 connection
	Arrived yesterday

	TBE2
	2029
	connection
	Arrived yesterday

	TBF1
	3059
	connection
	

	TBF2
	3034
	Cold tests
	


 In pipeline Next 1028

Test Coordinator this week:  Marco B;  MTM  Tech. on call:  (16-3602) : Guy D

· Situation From Cryogenics:  J-P Lamboy or ACR Rep.

· Other General Issues
1. Care in Ohmic M’ments 

2. Magnet arrivals & label + Notify Equip. Support

3. New OP staff arrivals ( 2 India, 3 AB-OP)

4. AOB

SnapShot of STATUS NOW(Tues 18/11 1700 hrs):

	SM18 LHC Magnet Test Facility [image: image1]


	TBA1
	_
	_
	 


	TBA2
	MBAR3022
	PT7.1 Minimum NRJ
	1.9 K


	TBB1
	_
	_
	 


	TBB2
	_
	_
	 


	TBC1
	MBAR3027
	PT12 Warm Up
	Warming


	TBC2
	SSS004
	PT12 Warm Up
	Warming


	TBD1
	_
	_
	 


	TBD2
	_
	_
	 


	TBE1
	MBAR2027
	New Magnet arrived
	300 K


	TBE2
	MBAR2029
	New Magnet arrived
	300 K


	TBF1
	MBAL3059
	ICS 2 Traveller AFTER Bolting
	300 K


	TBF2
	MBBR3034
	PT7.1 Training 1
	1.9 K
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Problems of Last week

	Date
	Problem
	Type
	Application/soft.
	Action/advise

	Tuesday 11_N


	TBE2: Tcoils problem

Bridge needed

IAP @ cold: error for D1_L-U (119 mV)


	MTM
	
	Federico solved the problem

Difficult to make better.

	Tuesday 11_N


	TBE2: PLCSE  OK  lost during ramp to 1 kA

Slow PA has to be repeated.


	MTM
	
	

	Wednesday 12_M


	TBE2: Impossible to put current
	AB/PO
	
	Power converter not working . It has been changed,

 

	Wednesday 12_M


	TBE2: MM0: MUX problem during auto-test
	MTM
	
	GPIB interface cable damaged.

 

	Wednesday 12_A


	TBE2: Provoked quench LF

Timing error (tfire~3s)
	MTM
	AQA
	 

	Thursday 13_A


	TBF2: Anticryostat (long) coefficient not correct. D07 instead of D11/12
	MTM
	
	Solved by Federico

 

	Thursday 13_N


	TBE2: Low charging rate of QH (LF)
	MTM
	
	 

	Thursday 13_N


	TBE1: Waiting for Cryo OK?
	ACR
	
	2 h  between 4.2 K and 1.9 K is foreseen (JP lamboy)

 

	Friday 14_M


	TBC1: IAP @ warm (prep) and HV test not done and magnet under pumping.
	MTM/ICS/ACR
	
	HV test done during the pumping as per instruction of test coordinator.

Traveller not updated.

 

	Friday 14_A


	TBA2: Transfer file of data for HV not done although message: transfer successfully done
	MTM
	HV
	Space problem.

Need space for the files.

 

	Friday 14_A


	Liquefactor problem : no cooling
	ACR
	
	 

	Friday 14_A


	TBE1: IAP@cold QH not discharged.


	ACR
	
	Wait for cryo OK.

 

	Saturday 15_A


	TBF2: Impossible to cool down 3034 from 4.5 K to 1.9 K
	ACR 
	
	Piquet cryo called.

 

	Saturday 15_N


	TBC1: Cryo problem with current leads.

Tprobe not working
	ACR 
	
	Solved by Bruno Sunday 16.

 

	Saturday 15_N


	TBF2: During selection of tests in local (IAP, MM0) Interconnection switch box off.

Trigger sent. Discharge of the heaters.
	MTM
	
	 

	Saturday 15_N


	TBF2: MM0 auto test.

Serial communication error for motor and amplifier.
	MTM
	
	 

	Sunday16_M

Sunday 16_A


	TBF2: Power Converter problem.

“Water fault”’
	MTM
	AB/PO
	Corrected Monday morning by AB/PO increasing the flow to 8 l/sec

 

	Sunday16_M


	TBE1: main rotary switch blocked, difficult to manipulate. 

Done by Federico and faulty signal (mechanical disalignment) 
	MTM
	
	 

	Sunday16_M


	TBA2: A2 C anticryostat goving –3.5 C 

PLC art not heating.
	ACR/MTM
	
	Solved.

 

	Sunday16_M


	TBE1: Misunderstanding with ACR team for RRR measurements.
	ACR/MTM
	
	Repeated later.

 

	Sunday16_A


	TBC1: Slow PA abort. “lead Temp.A-X1’’ problem
	ACR
	
	 

	Sunday16_N


	TBC1: Min energy 3 kA : trigger in Vbus ext.

Threshold in the card too low (1.16 V instead 1.6 V).
	MTM
	
	Corrected by Federico.

Repeated then OK

 

	Monday 17_M


	TBC1: MM0 No meas. Current in Ap1.

Cable used for TBC2.
	MTM
	
	Solved by Guy.

 

	Monday 17_M


	Computer T7 blocked.
	
	
	PC support called.

 

	Monday 17_A


	TBA2: Provoked quench all HF

Error in HF quench heater acquisition
	MTM
	AQA
	 

	Monday 17_A


	TBC1; PConverter ; interlock water
	AB/PO
	
	 


